Author's response to reviews

Title: All-cause and Cardiovascular mortality among ethnic German immigrants from the Former Soviet Union: a cohort study

Authors:

Ulrich Ronellenfitsch (ulrich.ronellenfitsch@urz.uni-heidelberg.de)
Catherine Kyobutungi (c.kyobutungi@urz.uni-hd.de)
Heiko Becher (heiko.becher@urz.uni-heidelberg.de)
Oliver Razum (oliver.razum@uni-bielefeld.de)

Version: 3 Date: 13 December 2005

Author's response to reviews:

Dear editor,

we thank you for having considered our paper for publication after the completed peer review process. Once again, we want to express our gratitude to both reviewers for their extremely helpful comments.

With regard to the second comment of Ms Miller, we appreciated her remark that the manuscript was in need of some grammatical revision. However, without being told concrete mistakes, it was not easy for us to notice which sentences were grammatically incorrect. We thoroughly revised the whole paper with regard to its grammar and changed the wording of some sentences. After that, we hope that now the manuscript meets the high language standards of your journal and that it is ready for publication.

Yours sincerely,

Ulrich Ronellenfitsch
Catherine Kyobutungi
Heiko Becher
Oliver Razum